Author’s response to reviews

Title: Genoprotective, antioxidant, antifungal and anti-inflammatory evaluation of hydroalcoholic extract of wild-growing Juniperus communis L. (Cupressaceae) native to Romanian southern sub-Carpathian hills

Authors:

Irina Fierascu (dumitriu.irina@yahoo.com)
Camelia Ungureanu (ungureanucamelia@gmail.com)
Sorin Marius Avramescu (sorin_avramescu@yahoo.com)
Carmen Cimpeanu (carmencimpeanu@yahoo.com)
Mihaela Ioana Georgescu (mihaelaigerogescu@yahoo.com)
Radu Claudiu Fierascu (radu_claudiu_fierascu@yahoo.com)
Alina Ortan (alina_ortan@hotmail.com)
Anca Nicoleta Sutan (ancasutan@yahoo.com)
Valentina Anuta (vali_anuta@yahoo.com)
Anca Zanfirescu (zanfirescuanca@yahoo.com)
Cristina Elena Dinu-Pirvu (ecristinaparvu@yahoo.com)
Bruno Stefan Velescu (bruno_velescu@yahoo.co.uk)

Version: 2 Date: 15 Dec 2017

Author’s response to reviews:

Response to Reviewer’s Comments

Dear Editor in Chief,

BMC Complementary and Alternative Medicine

Ms. Ref. No.: BCAM-D-17-00846R1
Title: Genoprotective, antioxidant, antifungal and anti-inflammatory evaluation of hydroalcoholic extract of wild-growing Juniperus communis L. (Cupressaceae) native to Romanian southern sub-Carpathian hills

We thank the reviewers and editor for their thoughtful comments regarding our manuscript. We took into account all of the comments and corrected our paper accordingly; all the changes are described below.

Editor: Please make the following revisions to your manuscript:

1. Please include the email addresses of all authors on the title page of your manuscript.
Answer: All authors email addresses were inserted in the manuscript, under a section called “Authors email addresses”: Page 1 – rows 22-23 and Page 2 – rows 1-4;

2. Please change the text under Consent to Publish to: "not applicable".
Answer: the text was changed – Page 27, row 6;

3. Please do not include the figure captions within the figure files or within the main manuscript, but instead place them at the end of the main manuscript in a section titled “Figure legends”.
Answer: The figure captions were removed from the main manuscript and placed at the end of the manuscript, in a section titled “Figure legends”: Page 35 rows 11-22, page 36 rows 1-2

Answer: the previous paper was already cited (reference 6); it was also inserted as a reference for table 2: Page 15, row 2.

Thank you very much for your appreciations and for your suggestions!

With regards

The authors